Author’s response to reviews

Title: Lipoprotein lipase in hemodialysis patients: indications that low molecular weight heparin depletes functional stores, despite low plasma levels of the enzyme

Authors:

Birgit Nasstrom (birgit.nasstrom.us@vll.se)
Bernd Stegmayr (bernd.stegmayr@medicin.umu.se)
Gunilla Olivecrona (gunilla.olivecrona@medbio.umu.se)
Thomas Olivecrona (thomas.olivecrona@medbio.umu.se)

Version: 3 Date: 22 October 2004

Author’s response to reviews:

Dear Editor
The reviewer is correct to insist that we change the title of our paper. He and we agree that our data strongly indicate that tissue stores are depleted but we have not (and can not) measured the tissue stores directly. The title now reads Indications that .... instead of a direct statement.
We have also made some small changes of language that do not affect the data or conclusions.
We have made the formatting changes you suggested. We had earlier submitted papers to BMC Physiology and to BMC Cell Biology and were surprised to find that the instructions for BMC Nephrology differ (e.g. Methods before Results rather than after Discussion.
We hope the manuscript is now ready to be accepted
Sincerely yours Thomas Olivecrona